Author's response to reviews

Title: A simple clinical model for planning transfusion quantities in heart surgery

Authors:

Felicetta Simeone (drfelix@libero.it)
Federico Franchi (franchif@unisi.it)
Gabriele Cevenini (cevenini@unisi.it)
Antonino Marullo (marullo@unisi.it)
Vittorio Fossumbron (v.fossumbron@ao_siena.toscana.it)
Sabino Scolletta (scolletta@unisi.it)
Bonizella Biagioli (biagiolib@unisi.it)
Pierpaoio Giomarelli (giomarelli@unisi.it)
Paolo Barbini (barbini@biolab.med.unisi.it)

Version: 5 Date: 23 May 2011

Author's response to reviews: see over
COVER LETTER

Editorial request: In your manuscript, please state the name of the local ethics committee that approved your study.

Reply: We have addressed the Editorial Request (see page 5, changes in yellow).

Request of referee 1: The authors responded very well to all the inquires except the Criticism 9. (i.e. Definitions of the three groups still different between the text and caption of Table 5)

Reply: We changed the caption of Table 5 (highlighted in yellow).